NBER WORKING PAPER SERIES

UNDERSTANDING THE ROLE OF GENETIC HETEROGENEITY IN SMOKING INTERVENTIONS: EXPERIMENTAL EVIDENCE FROM THE LUNG HEALTH STUDY

Shubhashrita Basu Jason Fletcher Qiongshi Lu Jiacheng Miao Lauren L. Schmitz

Working Paper 33473 http://www.nber.org/papers/w33473

NATIONAL BUREAU OF ECONOMIC RESEARCH 1050 Massachusetts Avenue Cambridge, MA 02138 February 2025

We thank participants from the following conferences for their helpful comments: Integrating Genetics and the Social Sciences (IGSS) 2022 conference, the October 2022 Virtual Seminar on the Economics of Risky Health Behaviors (VERB), and the Bringing Together Genetic Data with Economics (BRIDGE) 2023 conference. Funding for this project was generously provided by the National Institute on Aging (NIA) (R00 AG056599 and P30 AG017266). The content is solely the responsibility of the authors and does not necessarily represent the official views of the NIA. The Lung Health Study (LHS) was conducted and supported by the National Heart Lung and Blood Institute (NHLBI). This manuscript was prepared using LHS Research Materials obtained from the NHLBI Biologic Specimen and Data Repository Information Coordinating Center and does not necessarily reflect the opinions or views of the LHS or the NHLBI. This study was approved by the University of Wisconsin-Madison Institutional Review Board (IRB) (approval number: 2014-1350). The authors declare that they have no relevant or material financial interests that relate to the research described in this paper. The views expressed herein are those of the authors and doe not necessarily reflect the views of the National Bureau of Economic Research.

NBER working papers are circulated for discussion and comment purposes. They have not been peer-reviewed or been subject to the review by the NBER Board of Directors that accompanies official NBER publications.

© 2025 by Shubhashrita Basu, Jason Fletcher, Qiongshi Lu, Jiacheng Miao, and Lauren L. Schmitz. All rights reserved. Short sections of text, not to exceed two paragraphs, may be quoted without explicit permission provided that full credit, including © notice, is given to the source.

Understanding the Role of Genetic Heterogeneity in Smoking Interventions: Experimental Evidence from the Lung Health Study Shubhashrita Basu, Jason Fletcher, Qiongshi Lu, Jiacheng Miao, and Lauren L. Schmitz NBER Working Paper No. 33473 February 2025 JEL No. 11, 110, 118, J10

ABSTRACT

We examine whether genome-wide summary measures of genetic risk known as polygenic indices (PGIs) provide new insights into the efficacy of the Lung Health Study (LHS)–a large, randomized controlled trial (RCT) that evaluated the effect of a smoking cessation intervention program on cessation maintenance and lung function. Results indicate that the intervention was less successful for participants with higher PGIs for smoking initiation and intensity. Given the increasing availability and affordability of genomic data, we argue that in the context of RCTs, PGIs can further our understanding of heterogeneous treatment effects and the mechanisms that may be driving them.

Shubhashrita Basu Southern Utah University basu.shubhashrita@gmail.com

Jason Fletcher University of Wisconsin-Madison La Follette School of Public Affairs 1225 Observatory Drive Madison, WI 53706 and NBER jfletcher@lafollette.wisc.edu

Qiongshi Lu University of Wisconsin at Madison 425 Henry Mall #2104, Biotechnology Center Madison, WI 53706 qlu@biostat.wisc.edu Jiacheng Miao University of Wisconsin-Madison jmiao24@wisc.edu

Lauren L. Schmitz University of Wisconsin-Madison llschmitz@wisc.edu

A data appendix is available at http://www.nber.org/data-appendix/w33473

Understanding the Role of Genetic Heterogeneity in Smoking Interventions: Experimental Evidence from the Lung Health Study

Shubhashrita Basu[†], Jason Fletcher, Qiongshi Lu, Jiacheng Miao, Lauren L. Schmitz[†]

Abstract

We examine whether genome-wide summary measures of genetic risk known as polygenic indices (PGIs) provide new insights into the efficacy of the Lung Health Study (LHS)–a large, randomized controlled trial (RCT) that evaluated the effect of a smoking cessation intervention program on cessation maintenance and lung function. Results indicate that the intervention was less successful for participants with higher PGIs for smoking initiation and intensity. Given the increasing availability and affordability of genomic data, we argue that in the context of RCTs, PGIs can further our understanding of heterogeneous treatment effects and the mechanisms that may be driving them.

JEL Classification: 110, J10, J13

Keywords: Randomized controlled trial, heterogeneous treatment effects, polygenic index, smoking behavior, smoking cessation

[†]Corresponding authors. Basu: Department of Economics, Southern Utah University (shubhashritabasu@suu.edu); Fletcher: La Follette School of Public Affairs, University of Wisconsin-Madison (jason.fletcher@wisc.edu); Lu: Department of Biostatistics and Medical Informatics, University of Wisconsin-Madison (qlu@biostat.wisc.edu); Miao: Department of Biostatistics and Medical Informatics, University of Wisconsin-Madison (jmiao24@wisc.edu); Schmitz: La Follette School of Public Affairs, University of Wisconsin-Madison (llschmitz@wisc.edu).

We thank participants from the following conferences for their helpful comments: Integrating Genetics and the Social Sciences (IGSS) 2022 conference, the October 2022 Virtual Seminar on the Economics of Risky Health Behaviors (VERB), and the Bringing Together Genetic Data with Economics (BRIDGE) 2023 conference. Funding for this project was generously provided by the National Institute on Aging (NIA) (R00 AG056599 and P30 AG017266). The content is solely the responsibility of the authors and does not necessarily represent the official views of the NIA. The Lung Health Study (LHS) was conducted and supported by the National Heart Lung and Blood Institute (NHLBI). This manuscript was prepared using LHS Research Materials obtained from the NHLBI Biologic Specimen and Data Repository Information Coordinating Center and does not necessarily reflect the opinions or views of the LHS or the NHLBI. This study was approved by the University of Wisconsin-Madison Institutional Review Board (IRB) (approval number: 2014-1350). The authors declare that they have no relevant or material financial interests that relate to the research described in this paper.

I. Introduction

Although 20th-century declines in smoking prevalence have persisted, tobacco use is still the leading cause of preventable disease, disability, and death in the U.S., accounting for approximately one in five deaths (CDC 2024). The persistence of smoking-related morbidity and mortality is due in part to differences in the success of smoking cessation efforts within individuals and across population subgroups (Surgeon General 2020). A growing body of research suggests cessation outcomes may in part be driven by complex interactions between genetic and environmental (including therapeutic) influences (Surgeon General 2020; Fletcher 2012; Panagiotou et al. 2019).¹ However, the integration of genetic data into the evaluation of smoking cessation efforts has been limited, and additional research is needed to better understand whether precision medicine approaches that offer different treatments to smokers based on their genetics could be effective at helping people quit (Surgeon General 2020; Panagiotou et al. 2019).

To date, randomized controlled trials (RCTs) of smoking cessation with genetic information have been limited to the analysis of candidate genes—an approach that requires a prior hypothesis that a specific gene affects responsiveness to smoking cessation therapies or drugs. A meta-analysis of candidate gene RCTs identified some evidence that individuals carrying specific genotypes may be more likely to quit smoking with the use of nicotine replacement therapies but overall found no clear statistical evidence for differences by genotype (Panagiotou et al. 2019). However, candidate gene studies of complex phenotypes like smoking may also fail to replicate due to low power at the individual gene level and/or inflated false positives resulting from

¹ Twin and family studies have established that smoking behavior is moderately to highly heritable, with genetics explaining 46-84 percent of the variability in smoking initiation and smoking persistence, 75 percent of the variability in nicotine dependence, and 50-58 percent of the variability in smoking cessation (Broms et al. 2006; Vink et al. 2005; Xian et al. 2003).

publication bias (Duncan & Keller 2011).² Thus, research that uses other statistical methods for genetic risk stratification may be needed to improve understanding and guide pharmacotherapy choice for smoking cessation treatments.

In particular, the candidate gene approach differs from methods that aggregate variants across the genome to construct more powerful measures of genetic risk. As will be explained in more detail herein, results from a genome-wide association study (GWAS), which performs a hypothesis-free search for associations between a phenotype and millions of genetic variants, can be used to construct a polygenic index (PGI) or a weighted average of individual genetic predisposition for that phenotype.³ Because PGIs are objective, time-invariant, continuous measures that are more predictive than individual genetic variants, they may be beneficial for assessing heterogeneous treatment effects of programs or policies (Benjamin et al. 2012; Biroli et al. 2022). Moreover, in contrast to the traditional approach of identifying genetic risk based on subsets of the population with specific candidate genes, PGIs can identify underlying strata of disease risk in larger subsets of the population because they consider the cumulative impact of many common DNA variants spread across the genome (Fahed et al. 2022; Torkamani et al. 2018).

Additionally, given the low and continually falling cost of genomic profiling, PGIs may provide a cost-effective means to analyze biological pathways or mechanisms that drive treatment effects. Currently, for about \$40 per person, researchers can profile common genetic variation and

² Individual genetic loci influencing the etiology of complex phenotypes like smoking have low penetrance, meaning no single gene produces a symptom or trait at a detectable level (Gibson 2012).

³ PGIs have also been referred to in the literature as genetic risk scores (GRS), polygenic risk scores (PRS), or polygenic scores (PGS). We follow research that advocates for PGI terminology because it 1) can apply to clinical and behavioral outcomes in cases where a higher index does not correspond to increasing risk (e.g., educational attainment or subjective wellbeing), and 2) ethically, "index" has been viewed as more neutral compared to "score", which may give the false impression that a value judgment is being placed on individuals with higher or lower PGIs (Becker et al. 2021).

construct multiple PGIs for different phenotypes without incurring additional costs. Economists are increasingly utilizing PGIs in quasi-experimental studies to assess heterogeneous treatment effects from policy interventions (Schmitz and Conley 2017; Barcellos, Carvalho, and Turley 2018; Biroli et al. 2020; Fletcher and Lu 2021; Slob and Rietveld 2021). However, in the context of RCTs, PGIs remain underutilized despite their potential to enhance our understanding of how genetic diversity influences the estimation and interpretation of program effects.

In this study, we examine whether PGIs can further our understanding of treatment effect heterogeneity in a smoking cessation RCT using data from the Lung Health Study (LHS). To our knowledge, this is among the first studies in economics to use experimental variation to investigate genetic heterogeneity in treatment effects. The LHS was a large, multi-center RCT that evaluated the effect of a smoking cessation intervention program on cessation maintenance and lung function in 5,887 middle-aged smokers with early signs of chronic obstructive pulmonary disease (COPD). It was a relatively successful intervention that has been well-studied by clinicians and economists. The RCT had two treatment arms and a control arm. Both treatment groups received a smoking cessation intervention that combined behavior modification, the use of nicotine gum, and a fiveyear maintenance program to minimize relapse. In addition, the second treatment arm (SIA group) received a bronchodilator inhaler with the active drug ipratropium bromide (i.e., Atrovent) to assess whether it could further slow the rate of deterioration in lung function. In the five years following the initial intervention, 22 percent of treated participants had sustained quit rates compared with only five percent of control participants and showed significantly smaller declines in lung function (Anthonisen et al. 1994) and respiratory illness (Kanner and Anthonisen 2001). Downstream, the intervention reduced mortality in the treatment group by 15 percent (Anthonisen et al. 2005). In economics, research has explored the spillover impacts of the program, including the effects of treatment intervention on spousal quit rates (Fletcher & Marksteiner 2017), increases in body mass index (BMI) (Courtemanche, Tchernis, and Ukert 2018), and improvements in mental health (Meckel and Rittenhouse 2022).

Here, we utilize genotyped data from 4,145 participants to assess whether the success of the intervention depended on PGIs for smoking initiation (SI), smoking cessation (SC), and intensity, or cigarettes per day (CPD). We found the program was less successful for participants with a higher load of genetic variants that predisposed them to smoking addiction. Specifically, although both treatment arms were 23 percentage points (ppt) more likely to quit smoking on average, the success of the intervention declined by approximately 2.5 ppt for every one standard deviation (SD) increase in the SI PGI. However, although sustained quit rates were lower in higher SI PGI participants, the treatment did reduce genetic disadvantage in smoking intensity for individuals in the SIA group, regardless of quit status. A one SD increase in the CPD PGI, which in part captures genetic predisposition to faster nicotine metabolism, reduced CPD by one additional cigarette from the average treatment effect (ATE) of nine cigarettes. Our estimates are robust to additional controls and their interaction with the treatment and cannot be replicated using observed variation in baseline values for CPD.

Further, we show that PGIs can provide insights into mechanistic pathways at the biological or behavioral level that may be driving treatment effects. In particular, our results for smoking cessation and intensity appear to be driven by genetic pathways implicated in nicotine addiction and metabolism, as opposed to other potential smoking-adjacent PGIs for body mass index (BMI), lung function, and depression.⁴ We argue that the interpretation of treatment

⁴ For example, Cawley, Dragone, and Scholder (2016) explore smoking as a weight control mechanism and suggest that part of cigarette demand may be driven by efforts to manage weight. Thus, the genetics of BMI may also play a role in understanding pathways that influence treatment success.

differences by these PGIs is narrower compared to other demographic or socioeconomic characteristics and by comparing across PGIs, we can better understand the degree to which different dimensions of genetic risk contribute to the success of smoking interventions. Conversely, when we interact the treatment with characteristics like education, sex, or marital status, as is common in economics literature, we observe differences that could have multiple interpretations. For example, educational differences could capture not only genetic differences but also variations in social norms, responsiveness to health risk information, financial resources, and/or time preferences (Lillard 2017; Galama, Lleras-Muney, and Van Kippersluis 2018). Additionally, we find suggestive evidence that differential results by PGI were more pronounced in the subset of treated individuals who received the bronchodilator in addition to the smoking cessation intervention. While further research is needed, this suggests additional pharmacogenomic or therapeutic effects of ipratropium that were not identified in the original study, which did not find any benefits from inhaler use.

The rest of the paper is structured as follows. Section 2 describes the LHS intervention, data, and PGI construction in more detail. Section 3 describes our empirical strategy. Section 4 presents our results, and Section 5 concludes.

II. Data

A. The Lung Health Study Intervention

The LHS RCT was conducted from 1989 to 1994 in heavy smokers aged 35-60 who were diagnosed with early stages of chronic obstructive pulmonary disease (COPD) and who were motivated to quit.⁵ The study aimed to determine if a smoking cessation intervention and the use

⁵ Heavy smoking was defined as use of 10 cigarettes on at least one day during the prior 30 days. See O'Hara et al. (1993) for a detailed description of the LHS study design.

of a bronchodilator could slow the rate of decline in lung function. Participants had to be relatively healthy smokers at baseline, meaning they were excluded from the study if they had serious illnesses.⁶ A total of 5,887 participants were selected and randomized across ten participating clinical centers.

Eligible participants were randomly distributed into one of three groups: 1) smoking intervention plus inhaler with the active drug ipratropium bromide (Atrovent) (SIA); 2) smoking intervention plus placebo inhaler (SIP); or 3) no intervention (control group). Treated participants joined a 12-session group program over ten weeks that focused on cognitive and behavioral strategies for quitting and proper use of inhalers and nicotine gum (Connett et al. 1993; O'Hara et al. 1993). They had regular clinic visits and maintenance programs throughout the five-year follow-up. Additionally, treated participants received either an inhaler with ipratropium bromide (SIA group) or a placebo (SIP group) to assess if bronchodilator therapy slowed the rate of lung function decline. The inhaler, prescribed three times daily (two puffs/time), aimed to reduce airway hyperreactivity and improve airflow. Participants were encouraged to use the inhaler throughout the five-year trial, receiving a new supply at each four-month visit.

We utilize data from all five annual visits. Outcomes of interest include smoking cessation, which was validated by either salivary cotinine or carbon monoxide levels, and self-reported information on average CPD.

B. Polygenic Indices (PGIs)

We constructed PGIs using genotyped data that were collected from European ancestry participants with at least three time points of lung function data (N = 4,287) (Hansel et al. 2013).

⁶ See Online Appendix B2 for more specifics on the LHS intervention, including additional exclusionary criteria and program details.

LHS protocols did not exclude subjects based on race or ethnicity; however, only four percent of volunteers had non-European genetic ancestry backgrounds. GWAS are typically performed within ancestry groups because differences in allele frequency⁷ and linkage disequilibrium (LD) structure across populations distort estimated relationships in pooled samples, and estimates for one group are not necessarily accurate or valid for another.⁸ Although genotyped sample sizes for populations of non-European descent are beginning to reach sufficient power to produce separate GWAS of smoking behavior (e.g., Shrine et al. 2023), we are limited to the European ancestry subsample since genotype data on non-European ancestry participants were not available. After quality control (QC) and imputation of the genotyped data,⁹ 4,145 participants remained.¹⁰ Importantly, at baseline, the genotyped sample had similar demographics, BMI, and lung function as those who were not genotyped, and there is no evidence of selection into the sample by treatment

status (Online Appendix Tables A1-A2).

PGIs were constructed using coefficient estimates from GWAS. A GWAS meta-analyzes large samples of genetic data to test for associations between millions of single nucleotide polymorphisms (SNPs), or areas on the genome that vary between individuals, and a given phenotype of interest.¹¹ SNP effect sizes from GWAS were then used to construct a PGI for each

⁷ Allele frequency represents the incidence of a genetic variant in a population.

⁸ LD occurs when genetic variants are correlated with each other because of a lack of ancestral recombination events at locations in the DNA over time but could also arise from non-random mating in the population.

⁹ See Online Appendix B3 for details on QC and imputation of the genotyped data.

¹⁰ Genotyped data were merged with phenotype data based on several characteristics (a common identifier across the two samples was not available). After QC, 4,196 respondents out of 5,887 remained, and 51 could not be matched to the phenotypic data. Our final sample consists of 4,145 genotyped participants or 20,725 person-year observations across all five waves of the study.

¹¹ In the discovery phase, a GWAS will pool large consortia of genetic data using meta-analysis and run regressions testing each SNP at the genome-wide significance level of 5x10-8. In the replication phase, significant associations found in the discovery phase are tested in independent samples.

participant by multiplying the number of reference alleles a participant has at each SNP by the GWAS coefficient estimate for that SNP, and then aggregating these values across the genome to create a weighted scalar of genetic predisposition as follows:

$$PGI_i = \sum_{j=1}^n \beta_j \alpha_{ij} \tag{1}$$

Where *j* indexes the SNP, *i* indexes the individual, β is the coefficient estimate for SNP *j* from the GWAS, and α is the number of reference alleles (0, 1, or 2) for individual *i* at SNP *j*. PGIs were constructed using β estimates from GWAS available for SI, SC, and CPD (Liu et al. 2019). Additionally, PGIs for phenotypes with known biological ties to smoking behavior were constructed to examine other potential mechanistic pathways that may be driving treatment effects, including PGIs for BMI (Yengo et al. 2018), lung function (FEV₁/FVC ratio)¹² (Shrine et al. 2019), and depression (Nagel et al. 2018). Correlations between these PGIs are relatively weak, indicating limited genetic overlap (all *r*<0.20) (**Online Appendix Figure A1**). To avoid overfitting, we confirmed that the LHS sample was not included in the GWAS of interest.

To maximize prediction, PGIs were constructed using the software PRS-CS-auto, which uses a Bayesian framework to readjust SNP β weights from the GWAS to account for local LD patterns (Ge et al. 2019).¹³ All PGIs were standardized to have a mean of zero and a standard deviation of one for analysis. Higher PGIs for SI, SC, and CPD are associated with worse outcomes (i.e., a higher probability of becoming a regular smoker, a higher probability of not

 $^{^{12}}$ FEV₁/FVC is a ratio used in the diagnosis of lung diseases, particularly obstructive and restrictive lung conditions. It stands for forced expiratory volume in one second (FEV₁) over forced vital capacity (FVC) and is measured using spirometry. FEV₁ is the volume of air exhaled during the first second of forced exhalation and FVC is the total volume of air exhaled during the entire exhalation.

¹³ PRS-CS-auto models local LD patterns and uses a Bayesian regression framework with continuous shrinkage priors on SNP effect sizes to accommodate diverse underlying genetic architectures. It automatically learns the continuous shrinkage prior from the GWAS summary statistics, and no validation dataset is needed. We used the 1000 Genomes Project Phase 3 European samples as the LD reference (Auton et al. 2015).

quitting, and increases in CPD). Similarly, for BMI, depressive symptoms, and lung function, higher PGIs are associated with worse outcomes. In analyses with PGIs, we account for population stratification by controlling for the first 20 principal components (PCs) of the genetic data (Price et al. 2006).¹⁴

Online Appendix Table A3 reports the predictive power of the PGIs in the LHS sample, or the incremental R² from a regression of the phenotype on the PGI after conditioning on the first 20 PCs. The SC PGI explains 0.2 percent of the variance in quit status (across all five visits), and the CPD PGI explains 3.3 percent of the variance in CPD at baseline. The BMI PGI explains 8.8 percent of the variance in BMI at baseline, and the FEV₁/FVC PGI explains 0.01 percent of the variance in FEV₁/FVC at baseline. These estimates are on par with the predictive power of these PGIs in larger population-based studies (Yengo et al. 2018; Liu et al. 2019). Further, if we stratify participants by PGI quintile, we see a gradual increase in the outcome mean across quintiles of the CPD, BMI, FEV₁/FVC, and depression PGIs (**Online Appendix Figures A2-A5**).¹⁵

Not surprisingly, because the SI PGI was trained to predict the probability of becoming a regular smoker or smoking at least 100 cigarettes, it only explains 0.03 percent of the variation in smoking cessation. However, we include it in our analyses because it captures other biological pathways that are distinct from those implicated in the SC or CPD PGIs that could interact with the treatment, including pathways involved in dopaminergic and neurotransmission systems that

¹⁴ Although our analyses are limited to individuals of European ancestry, estimates still need to account for population stratification within ancestry that could confound associations between the PGI and the phenotype of interest. Population stratification arises when ancestral subpopulations are isolated geographically, with low rates of migration and gene flow throughout several generations (Hellwege et al. 2017). Cultural and environmental differences can also induce stratification, even if population subgroups inhabit the same geographical region (ibid). Additionally, assortative mating can produce population structure-related confounding effects (Young et al., 2019). PCs were calculated using PLINK software (Purcell et al. 2007).

¹⁵ Results are not presented for smoking cessation because we do not have baseline pre-treatment values for quit status.

affect long-term reward processing and addiction (Liu et al. 2019). As a point of comparison, in **Online Appendix Table A3**, we also report incremental R^2 estimates between our outcomes of interest and demographic characteristics that are often used to analyze heterogenous treatment effects of smoking interventions, including sex and marital status. The PGIs either explain a greater proportion of the variance in our outcomes of interest, or their explanatory power is not significantly different from these characteristics.

The predictive power of a PGI depends on the trait's heritability and the power of the GWAS from which it was derived (Wray et al. 2019). Although the predictive power of the PGIs in the LHS sample is not negligible, it is a fraction of SNP-based measures of heritability for these phenotypes, which range from approximately five percent for smoking cessation to eight percent for SI and CPD (Liu et al. 2019).¹⁶ Thus, the predictive power of these PGIs will likely increase as GWAS sample sizes continue to grow. However, because GWAS estimates exploit pooled samples of unrelated individuals, the predictive power of the PGI may also exceed what can be explained by direct genetic effects because estimates in unrelated individuals also capture indirect genetic effects and confounding from population stratification (Young 2019).¹⁷ Conditioning on the first 20 PCs of the genetic data accounts for any confounding from population stratification (Price et al. 2006). Thus, the remaining variation in the PGIs reflects both direct and indirect genetic effects. To isolate direct genetic effects, researchers are beginning to conduct within-

¹⁶ Heritability is often estimated using phenotype correlations between identical and non-identical twins, however GWAS data on unrelated individuals can also be used to estimate the phenotypic variation explained by SNPs on a genotyping array (Yang et al. 2017). These so-called SNP-based heritability estimates are typically lower than twinbased heritability estimates (Manolio et al. 2009). Some of this gap may be explained by rare genetic variants that are not captured on genotype arrays, which were designed to capture common genetic variation within ancestral populations.

¹⁷ Indirect effects may arise because a genetic variant's effect is working through other (indirect) biological or environmental pathways. For example, a variant may influence a trait via its effect on parental behavior.

family GWASs that exploit genetic data on parent-offspring trios and sibling pairs (Howe et al. 2022). However, because larger genotyped family samples are needed to reach sufficient power, these efforts are still in their infancy.

C. Summary Statistics

Table 1 compares baseline characteristics for the sample by treatment status. Mean values at baseline are comparable across groups. On average, participants were approximately 48.5 years old at the start of the trial (SD=6.7), 35-39 percent female (SD=0.48), and had 13.6 years of education (SD=2.8). Participants were long-term heavy smokers, having smoked an average of one pack per day for 40.7 years (SD=18.2) and were currently smoking an average of 31 CPD at baseline (SD=12.5). The average FEV₁/FVC ratio across all groups was 63 percent (SD=5.5), which is seven ppt below the normal value of 70 percent, indicating early signs of lung function decline. The average BMI was 25.5 (SD=3.9), which is comparable to the U.S. average in 1990 for that age group (Freedman et al. 2002). The PGIs, which were standardized to have a mean of zero and an SD of one in the full analysis sample, exhibit similar distributions across treatment and control subgroups.

Online Appendix Figure A6 shows the per-wave averages for smoking cessation and CPD from baseline through wave five by treatment status for the genotyped sample. Overall, the treatment was successful: quit rates were approximately 38 percent for both treatment arms across all waves, compared to an initial 9 percent for the control group, which increased to 23 percent by the fifth visit. Average CPD declines were sharper for the treatment group (20) compared to the control group (13).

III. Empirical Strategy

To examine whether treatment effects varied by PGI, we estimate the following equation:

$$Y_{ict} = \alpha + \beta Treated_i + \gamma PGI_i + \delta Treated_i \times PGI_i + \theta Female_i + PC'_i \pi + \rho_c + \sigma_t + \varepsilon_{it}$$
(2)

where Y_{ict} is the outcome of individual *i* observed at age *c* in visit *t*, *Treated*_{*i*} is an indicator for whether individual *i* was assigned to one of the treatment arms, *PGI*_{*i*} is the polygenic index of interest for *i*, and *PC*_{*i*} is a matrix containing the first 20 PCs of the genetic data to account for population stratification. We also control for sex and include fixed effects for age (ρ_c) and visit (σ_t) to improve the precision of our estimates. Regressions are estimated separately for each treatment arm and both treatment groups combined. Standard errors are clustered at the individual level.

Our parameter of interest is δ , which captures the degree to which the average treatment effect (β) varies by participants' underlying PGI. If δ is significantly different from zero, it implies that the efficacy of the treatment was dependent on participants' underlying PGI. Models with continuous outcomes are estimated using ordinary least squares (OLS). For dichotomous degree outcomes, we report results estimated from linear probability models.

IV. Results

A. Heterogeneous treatment effects by PGI

Table 2 presents the main effects and heterogenous treatment effects for SC and CPD. Columns report results separately for both treatment arms combined (Composite) and for the SIA and SIP arms. Treated individuals were 23 ppt more likely to quit smoking, regardless of the treatment arm, which is comparable to results from the original trial. Since participants in the trial were already heavy smokers, we did not find a significant main effect of the SI PGI on quit status, which was trained to predict the probability of regular smoking relative to individuals who never smoked. However, we do find that the treatment effect was smaller for participants with higher SI PGIs: a one SD increase in the SI PGI reduced the likelihood of quitting by an additional 2.5 ppt for both treatment arms combined. Conversely, while there is a statistically significant main effect of the SC PGI on quit rates, the effect of the treatment does not vary by the SC PGI (**Online Appendix Table A4**). This could reflect stronger interactive effects between the intervention and genetic variants identified in the SI GWAS, which play a larger role in sustaining neuronal pathways that regulate reward processing and addiction, as opposed to pathways implicated in the SC and CPD PGIs, which are more involved in regulating nicotine metabolism.¹⁸

The magnitude and significance of the SI PGI interaction appear to be driven by individuals in the SIA arm, where we observe a 3.5 ppt statistically significant reduction in quit rates for each SD increase in the PGI compared to a statistically insignificant reduction of 1.3 ppt for participants in the SIP arm. This may suggest additional pharmacogenomic effects of the ipratropium inhaler that were not identified in the original study, which did not detect differences in cessation or FEV₁/FVC between the SIA and SIP treatment arms. Ipratropium is an antimuscarinic that reduces airway reactivity in smokers by blocking the muscarinic cholinergic receptors, which in turn both decreases the production of certain proteins that increase mucous production and dilates bronchial smooth muscle in the lungs. Importantly, antimuscarinics can act on muscarinic receptors in the brain and have been shown to decrease the intensity of cravings in patients recovering from cocaine use (Grasing 2016). Because the SI PGI is enriched for central nervous system functions related to reward processing and addiction, this may explain why we see a potential pharmacogenomic

¹⁸ For example, well-researched candidate genes that have been replicated in GWAS of CPD and SC include the nicotine metabolism gene *CYP2A6* and the *CHRNA5-A3-B4* gene cluster, which encodes subunits of nicotinic acetylcholine receptors (nAChR) (Liu et al. 2019). Conversely, genes implicated in SI are more related to the modulation of dopamine reward circuits that promote continued use of nicotine after initial exposure, including *BDNF* and *PPP1R1B*, which regulate neuronal plasticity that underlies reward-based learning (Furberg et al. 2010; Liu et al. 2019).

interaction with the SI PGI as opposed to the SC PGI. However, because estimates for the treatment arms are not significantly different from one another, this interpretation is suggestive and in need of further study.

Concerning smoking intensity, the treatment significantly reduced genetic disadvantage in CPD at the extensive margin, regardless of quit status: A one SD increase in the CPD PGI reduced CPD by one additional cigarette relative to the ATE of nine cigarettes. These results can also be seen in **Figure 1**, which plots the estimated treatment effects by PGI for quit status and CPD. The treatment increased genetic disparities in smoking cessation relative to the control group. In other words, the treatment was not as effective in curbing quit rates for participants with higher genetic risk; however, although higher PGI participants had more difficulties quitting, the intervention did appear to reduce genetic disparities in CPD such that by the end of the trial, treated participants with higher CPD PGIs were no longer smoking more CPD on average than participants with lower PGIs. Finally, the magnitude and significance of the CPD PGI interaction are stronger for participants in the SIA arm. However, like smoking cessation, these estimates are not statistically different between treatment arms.

B. Using PGIs to enhance understanding of treatment pathways

Examining the interplay between the treatment and other PGIs related to smoking behavior may provide additional clues on the mechanisms through which genetic factors are influencing treatment outcomes (Biroli et al. 2022; Barth, Papageorge, and Thom 2020). To assess whether our results are driven by smoking genetics as opposed to other genetic pathways that could influence smoking behavior, we added additional PGI interactions with the treatment to our empirical model (**Table 3**). These results also condition on interactions between the genetic PCs and the treatment. Importantly, we focused on PGIs that contain genetic variants with known pleiotropic effects¹⁹ on smoking or PGIs with biological ties to smoking behavior, including PGIs for BMI, FEV₁/FVC, and depression.²⁰ Of note, we do not include the educational attainment PGI in our analyses because past research has shown that its correlation with smoking PGIs for these birth cohorts is driven by the emergence of the educational gradient in smoking and is thus a byproduct of environmental forces as opposed to genetic differences in cognitive ability or risk preferences (Galama, Lleras-Muney, and Van Kippersluis 2018; Wedow et al. 2018;).²¹

Our results are robust to the inclusion of these additional PGIs, and their interactions with the treatment are smaller in magnitude and less precise than the SI and CPD PGI interaction effects. This suggests that differences in treatment success were operating primarily through the genetics of smoking as opposed to genetic pathways implicated in BMI, lung function, or depression PGIs.

C. Additional analyses and robustness checks

The magnitude of the PGI-by-treatment interaction effect is comparable to other heterogeneous treatment effects by gender, education, and marital status (**Table 4**). Being married increased the likelihood of smoking cessation for treated participants by 4.4 ppt, perhaps due to spousal spillover effects, which have been documented previously in the LHS (Fletcher and

¹⁹ Pleiotropic effects could manifest as biological or mediated pleiotropy. Biological pleiotropy occurs when a genetic variant has a direct or independent influence on more than one trait, whereas mediated pleiotropy occurs when a variant affects smoking behavior through its effect on another trait (Solovieff et al. 2013).

²⁰ Multiple variants identified in GWAS of BMI appear to play a role in both increased obesity and smoking, including variants in the *CHRNA5-A3-B4* gene cluster (e.g., Thorgeirsson et al. 2013; Wang et al. 2017). For lung function, research has identified positive genetic correlations between CPD, SC, and the risk of lung cancer and COPD such that genetic variants that alter smoking are also thought to contribute to genetic risk for lung cancer and COPD, including variants in the *CHRNA5-A3-B4* gene cluster and chromosomal region 19q13.2, which includes *CYP2A6*, the gene encoding the enzyme that metabolizes nicotine (e.g., Parker et al. 2019; Bray et al. 2020). Finally, studies support the existence of genetic correlation between SI and depressive symptoms (Bulik-Sullivan et al. 2015; Zheng et al. 2017). Although more research is needed, results suggest that bidirectional associations between depressive symptoms and SI may be partially accounted for by shared genetic factors (Schmitz, Gard, and Ware 2019).

²¹ This is consistent with prior research that suggests latent traits like intelligence play a limited role in explaining socioeconomic disparities in rates of smoking (Cutler and Lleras-Muney 2006; Pampel, Krueger, and Denney 2010).

Marksteiner 2017). Every one-year increase in education in the treated group increased the probability of cessation by 1.1 ppt. Treated female participants smoked 2.4 CPD more per day than treated males. There were no significant differences in CPD for treated participants by marital status or education.

As a counterpoint to the PGI estimates, we assessed whether we could draw similar conclusions for smoking cessation using pretreatment or baseline variation in smoking behavior (**Online Appendix Table A5** and **Figure A7**). Since quit status cannot be observed pre-treatment, we used baseline variation in CPD. While baseline CPD is negatively associated with lower quit rates post-treatment, we do not find evidence of differences in smoking cessation by baseline CPD. Additionally, our SI PGI interaction findings are robust to controls for baseline CPD and its interaction with the treatment. These results suggest the SI PGI is capturing unique pre-treatment variation in quit status that is not captured in observed smoking behavior at baseline.

Finally, we evaluated whether the average effect estimated across all visits varied in the short or medium term. Effects remain consistent across visits except for the main effect of the treatment on smoking cessation and CPD (**Online Appendix Tables A6-A7**). Here, the treatment effect decreases by approximately one-third to one-half for smoking cessation and CPD, respectively, but the effects of the PGIs and their interaction with the treatment remain stable across visits. This implies that the relative advantages (or disadvantages) conferred to participants from their genetics remained relatively stable throughout the trial.

V. Conclusion

To improve public health, more effective smoking cessation efforts are needed to reduce disparities in smoking-related morbidity and mortality. Understanding factors that contribute to disparities in treatment efficacy is crucial for evaluating existing policies and using data to forecast the success of new programs and interventions (Heckman 2001). This study leveraged advances in statistical genetics and data from a large smoking cessation RCT to assess whether the impact of the treatment varied by PGIs for smoking behavior. Results indicate that quit rates were lower for genetically at-risk participants. However, regardless of quit status, the treatment did reduce genetic disparities in smoking intensity such that treated participants with higher genetic risk were no longer smoking more CPD on average compared to their lower-risk counterparts.

This suggests the efficacy of cessation treatments may be improved by assigning participants to specific treatments based on the results of genetic testing. Likewise, focusing on participants who are more genetically responsive to interventions may increase the statistical power of RCTs by increasing the relative proportion of participants that are more responsive to the intervention, thereby decreasing sample size and cost while also increasing the benefits of the trial (Fahed, Philippakis, and Khera 2022). However, before pharmacogenetic stratification becomes routine, further clinical trials are needed, both to assess whether this approach improves overall cessation outcomes in a substantial way for participants in high-risk strata as well as to determine the cost-effectiveness of this approach (Surgeon General 2020).

Despite these promises, several limitations warrant discussion. First, the predictive power of PGIs is dependent on trait heritability and the power of the GWAS from which they were derived, which in turn may limit their utility in certain settings (Daetwyler, Villanueva, and Woolliams 2008). For example, certain behavioral outcomes of interest to economists may be difficult to apply to a GWAS setting because they are not available in large studies with genetic data or because they are empirically difficult to measure or define. Moreover, if the heritability of a trait is low, the PGI may not have adequate predictive power to quantify strata of genetic risk. Second, because PGIs capture phenotypic differences that arise from common DNA variants, they may misrepresent genetic risk in situations where rare variants play an outsize role in the onset or progression of a disease or trait (Lewis and Vassos 2020). Third, because GWAS meta-analyzes summary statistics across samples of unrelated individuals to achieve adequate sample size, they capture not only direct genetic effects but also indirect effects and confounding from population stratification (Young 2019). Although this presents a challenge for identifying causal variants, there is limited evidence in the literature regarding the extent to which these issues impact the use of PGIs for risk stratification and precision medicine. Fourth, because the majority of GWAS to date have been conducted in individuals of European ancestry, their predictive power may be limited in other ancestral groups (Martin et al. 2019).

However, we view these current challenges as temporary and empirically tractable as the size of genetic samples continues to increase dramatically due to persistent and rapid declines in the cost of whole genome sequencing. In particular, new statistical methods coupled with large-scale global efforts aimed at increasing the diversity of genetic samples are already reducing disparities in PGI prediction across ancestral populations (Ruan et al., 2022; Miao et al. 2023) and fueling the discovery of new disease-related variants (Gudmundsson et al. 2022). Increasing genetic samples in family and sib-based studies will improve the power of within-family GWAS and aid in decomposing direct and indirect genetic effects (Wu et al., 2021.; Howe et al. 2022). Finally, future improvements in algorithms and machine learning approaches that can incorporate other molecular and biomarker data alongside genetic data may outperform current PGI aggregation methods and improve our understanding of mechanistic pathways (Eldjarn et al. 2023).

Tables and Figures

	Control		SIP		SIA	
	Mean	SD	Mean	SD	Mean	SD
Baseline characteristics						
Age	48.471	6.727	48.694	6.741	48.450	6.681
Female	0.351	0.478	0.362	0.481	0.394	0.489
Pack years	40.731	18.244	40.338	18.173	40.682	19.481
Years of education	13.647	2.769	13.563	2.849	13.630	2.833
Cigarettes per day (CPD)	30.950	12.501	31.468	12.524	31.333	13.173
Married	0.706	0.456	0.714	0.452	0.704	0.457
Body mass index (BMI)	25.508	3.804	25.769	3.862	25.318	3.854
FEV ₁ /FVC	63.028	5.450	63.055	5.432	62.750	5.659
Polygenic indices (PGIs)						
Smoking initiation (SI)	-0.067	0.979	0.047	1.000	0.021	1.018
Smoking cessation (SC)	-0.005	0.983	0.013	1.043	-0.008	0.974
Cigarettes per day (CPD)	-0.024	0.970	-0.041	0.998	0.063	1.029
Body mass index (BMI)	-0.020	0.970	0.012	0.985	0.008	1.043
FEV ₁ /FVC	0.038	0.996	-0.011	1.002	-0.027	1.002
Depression	0.023	1.001	-0.027	0.994	0.003	1.005
Ν	1383		1355		1407	

Table 1. Mean baseline characteristics by treatment status for the LHS genetic sample

Note: The table presents the mean and standard deviation (SD) of baseline characteristics by treatment status for the LHS genetic sample. FEV_1/FVC : forced expiratory volume in one second (FEV_1) divided by forced vital capacity (FVC). SIA: smoking intervention plus Atrovent inhaler group. SIP: smoking intervention plus placebo inhaler group. Pack years are calculated by multiplying the number of cigarettes smoked per day by the number of years smoked. PGIs are standardized to have a mean of zero and an SD of one in the full sample.

	Composite	SIA	SIP		
Panel I	Smoking Cessation				
Treated	0.230	0.232	0.227		
	(0.011)	(0.014)	(0.014)		
SI PGI (std)	0.004	0.005	0.002		
	(0.008)	(0.008)	(0.008)		
Treated \times SI PGI (std)	-0.025	-0.035	-0.013		
	(0.011)	(0.013)	(0.013)		
Ν	20,725	13,950	13,690		
Panel II		CPD			
Treated	-9.072	-9 340	-8.742		
	(0.415)	(0.474)	(0.489)		
CPD PGI (std)	1.717	1.703	1.776		
	(0.370)	(0.371)	(0.370)		
Treated \times CPD PGI (std)	-0.618	-1.017	-0.204		
	(0.441)	(0.502)	(0.508)		
Ν	20,279	13,644	13,404		

Table 2. Heterogeneous treatment effects of the LHS intervention on smoking outcomes by PGIs for smoking initiation and cigarettes per day

Note: SI: smoking initiation; CPD: cigarettes per day. Composite indicates both treatment arms combined. SIA: smoking intervention plus Atrovent inhaler group. SIP: smoking intervention plus placebo inhaler group. PGIs are standardized. All regressions control for age fixed effects, wave fixed effects, sex, and genetic PCs 1-20. Standard errors are clustered at the individual level.



Figure 1. Treatment effects of the LHS intervention on smoking cessation and cigarettes per day by PGI

Standardized value of CPD PGI

Note: The figure shows predicted smoking cessation and cigarettes smoked per day (CPD) by standardized values of the smoking initiation (SI) and CPD PGIs, respectively. Composite indicates both treatment arms combined. SIA: smoking intervention plus Atrovent inhaler group. SIP: smoking intervention plus placebo inhaler group. Estimates are from regressions that control for age fixed effects, wave fixed effects, sex, and genetic PCs 1-20. Standard errors are clustered at the individual level. 95% confidence intervals are reported.

Smoking cessation Treated 0.229 0.230 0.227 (0.011) (0.014) (0.014) SLPCL (atd) 0.002 0.002 0.001	
Treated 0.229 0.230 0.227 (0.011) (0.014) (0.014) SLPCL (std) 0.002 0.002	
$(0.011) \qquad (0.014) \qquad (0.014)$	
SLDCL(atd) = 0.002 = 0.001	
SI PGI (std) 0.002 0.002 0.001	
$(0.008) \qquad (0.008) \qquad (0.008)$	
Treated × SI PGI (std) -0.020 -0.029 -0.010	
(0.011) (0.014) (0.014)	
CPD PGI (std) -0.014 -0.015	
(0.008) (0.008) (0.008)	
Treated × CPD PGI (std) $0.012 0.032 -0.007$	
(0.012) (0.014) (0.014)	
BMI PGI (std) 0.004 0.004 0.004 (0.000)	
(0.009) (0.009) (0.009)	
Ireated × BMI PGI (std) -0.012 -0.018 -0.009 (0.012) (0.014) (0.015)	
(0.012) (0.014) (0.013)	
$\begin{array}{cccc} FEV_{1}/FVC PGI (std) & 0.003 & 0.003 & 0.004 \\ (0.009) & (0.009) & (0.009) & (0.009) \end{array}$	
(0.009) (0.009) (0.009)	
$\begin{array}{cccc} \text{Treated} \times \text{FEV}_{1}/\text{FVC} \text{FOT}(\text{std}) & 0.000 & 0.001 & -0.001 \\ (0.012) & (0.014) & (0.014) \end{array}$	
Depression PGI (std) = 0.002 = 0.002 = 0.002	
$\begin{array}{c} 0.005 \\ (0.008) \\ ($	
(0.000) (0.000) (0.000)	
$\begin{array}{cccc} \text{Treated} \times \text{Depression PGI}(\text{std}) & -0.012 & -0.017 & -0.007 \\ (0.012) & (0.014) & (0.014) \\ \end{array}$	
(0.012) (0.014) (0.014)	
N 20,725 13,950 13,690	
CPD	
Treated -9.066 -9.260 -8.805	
(0.424) (0.485) (0.497)	
SI PGI (std) 0.019 0.015 0.045	
(0.340) (0.340) (0.339)	
Treated \times SI PGI (std)0.4040.6990.086	
(0.419) (0.480) (0.487)	
CPD PGI (std) 1.820 1.815 1.845	
(0.372) (0.373) (0.371)	
Treated × CPD PGI (std) -0.796 -1.329 -0.307	
(0.449) (0.515) (0.519)	
BMI PGI (std) -0.245 -0.245 -0.243	
(0.363) (0.364) (0.363)	
Treated × BMI PGI (std) 0.142 0.318 0.034 (0.427)(0.404)(0.500)	
(0.437) (0.494) (0.506)	
$\begin{array}{cccc} FE v_{1}/F v C PGI (sta) & 0.254 & 0.253 & 0.216 \\ (0.349) & (0.351) & (0.350) \end{array}$	

Table 3. Heterogeneous treatment effects of the LHS intervention on smoking outcomes by PGIs for smoking, body mass index, lung function, and depression

$Treated \times FEV_{l}/FVC \ PGI \ (std)$	0.192	0.075	0.362
	(0.430)	(0.497)	(0.497)
Depression PGI (std)	-0.268	-0.267	-0.258
	(0.348)	(0.349)	(0.348)
Treated \times Depression PGI	0.726	0.919	0.534
	(0.430)	(0.490)	(0.507)
Ν	20.279	13,644	13,404

Note: Composite indicates both treatment arms combined. SIA: smoking intervention plus Atrovent inhaler (ipratropium bromide) group; SIP: smoking intervention plus placebo inhaler group. FEV₁/FVC: forced expiratory volume in one second (FEV₁) over forced vital capacity (FVC). PGIs are standardized. All regressions control for age fixed effects, wave fixed effects, sex, genetic PCs 1-20, and genetic PCs 1-20 interacted with the treatment. Standard errors are clustered at the individual level.

· · · · · ·	Composite	SIA	SIP
—	1	Smoking Cessation	
Treated	0.054	-0.015	0.139
	(0.064)	(0.076)	(0.080)
SI PGI (std)	0.005	0.005	0.003
	(0.008)	(0.008)	(0.008)
Treated \times SI PGI (std)	-0.023	-0.031	-0.012
	(0.011)	(0.013)	(0.013)
Married	0.013	0.016	0.012
	(0.018)	(0.018)	(0.018)
Treated \times Married	0.044	0.074	0.009
	(0.025)	(0.030)	(0.031)
Years of Education	0.005	0.005	0.004
	(0.003)	(0.003)	(0.003)
Treated \times Years of Education	0.011	0.015	0.006
	(0.004)	(0.005)	(0.005)
Female	-0.010	-0.010	-0.009
	(0.017)	(0.017)	(0.017)
Treated \times Female	-0.016	-0.017	-0.015
	(0.024)	(0.029)	(0.030)
Ν	20,720	13,945	13,685
		CPD	
Treated	-7.856	-6.844	-8.996
	(2.360)	(2.648)	(2.766)
CPD PGI (std)	1.699	1.689	1.735
	(0.364)	(0.365)	(0.365)
Treated \times CPD PGI (std)	-0.695	-1.072	-0.266
	(0.437)	(0.496)	(0.509)
Married	-2.838	-2.922	-2.848
	(0.782)	(0.780)	(0.784)
Treated × Married	-0.188	-0.515	0.283
	(0.962)	(1.102)	(1.121)
Years of Education	-0.266	-0.265	-0.243
	(0.125)	(0.125)	(0.125)
Treated × Years of Education	-0.140	-0.244	-0.032
	(0.149)	(0.166)	(0.176)
Female	-3.627	-3.616	-3.657
	(0.698)	(0.697)	(0.698)
Treated × Female	2.357	3.336	1.373
	(0.851)	(0.973)	(0.988)
Ν	20,274	13,639	13,399

Table 4. Heterogeneous treatment effects of the LHS intervention on smoking outcomes
by smoking PGIs and demographic characteristics

Note: Composite indicates both treatment arms combined. SIA: smoking intervention plus Atrovent inhaler group. SIP: smoking intervention plus placebo inhaler group. PGIs are standardized. All regressions control for age fixed effects, wave fixed effects, sex, and genetic PCs 1-20. Standard errors are clustered at the individual level.

References

- Anthonisen, N., J. Connett, J. Kiley, M. Altose, W. Bailey, W. Buist, W. Conway, P. Enright, et al. 1994. "Effects of Smoking Intervention and the Use of an Inhaled Anticholinergic Bronchodilator on the Rate of Decline of FEV1." *JAMA* 272 (19): 1497-505. https://pubmed.ncbi.nlm.nih.gov/7966841/
- Anthonisen, N., M. Skeans, R. Wise, J. Manfreda, R. Kanner, and J. Connett. 2005. "The Effects of a Smoking Cessation Intervention on 14.5-Year Mortality." *Annals of Internal Medicine* 142: 233–239. <u>https://pubmed.ncbi.nlm.nih.gov/15710956/</u>
- Auton, Adam, Gonçalo R. Abecasis, David M. Altshuler, Richard M. Durbin, David R. Bentley, Aravinda Chakravarti, Andrew G. Clark, et al. 2015. "A Global Reference for Human Genetic Variation." *Nature*. Nature Publishing Group. <u>https://doi.org/10.1038/nature15393</u>
- Barcellos, Silvia H., Leandro S. Carvalho, and Patrick Turley. 2018. "Education Can Reduce Health Differences Related to Genetic Risk of Obesity." *Proceedings of the National Academy of Sciences of the United States of America* 115 (42): E9765–72. <u>https://doi.org/10.1073/pnas.1802909115</u>
- Barth, D., N. Papageorge, and K. Thom. 2020. "Genetic Endowments and Wealth Inequality." Journal of Political Economy 128 (4). <u>https://www.journals.uchicago.edu/doi/abs/10.1086/705415</u>
- Becker, Joel, Casper A.P. Burik, Grant Goldman, Nancy Wang, Hariharan Jayashankar, Michael Bennett, Daniel W. Belsky, et al. 2021. "Resource Profile and User Guide of the Polygenic Index Repository." *Nature Human Behaviour* 5 (12): 1744–58. https://doi.org/10.1038/s41562-021-01119-3
- Benjamin, Daniel J., David Cesarini, Christopher F. Chabris, Edward L. Glaeser, David I. Laibson, Vilmundur Gunason, Tamara B. Harris, et al. 2012. "The Promises and Pitfalls of Genoeconomics." *Annual Review of Economics* 4 (July):627–62. <u>https://doi.org/10.1146/annurev-economics-080511-110939</u>
- Biroli, Pietro, Titus J. Galama, Stephanie von Hinke, Hans van Kippersluis, Cornelius A. Rietveld, and Kevin Thom. 2022. "The Economics and Econometrics of Gene-Environment Interplay," March. <u>http://arxiv.org/abs/2203.00729</u>
- Biroli, P., C. Zünd, J. Abeler, S. Barcellos, D. Benjamin, L. Carvalho, L. Cesarini, et al. 2020."Genes, Pubs, and Drinks: Gene-Environment Interplay and Alcohol Licensing Policy in the UK." *Working Paper*.

https://www.mri.cornell.edu/sites/default/files/PAM/HEHBAD/Papers/genexpub.pdf

Bray, Michael J., Li Shiun Chen, Louis Fox, Dana B. Hancock, Robert C. Culverhouse, Sarah M. Hartz, Eric O. Johnson, et al. 2020. "Dissecting the Genetic Overlap of Smoking Behaviors, Lung Cancer, and Chronic Obstructive Pulmonary Disease: A Focus on Nicotinic Receptors and Nicotine Metabolizing Enzyme." *Genetic Epidemiology* 44 (7): 748–58. <u>https://doi.org/10.1002/gepi.22331</u>

- Broms, Ulla, Karri Silventoinen, Pamela A. F. Madden, Andrew C. Heath, and Jaakko Kaprio. 2006. "Genetic Architecture of Smoking Behavior: A Study of Finnish Adult Twins." *Twin Research and Human Genetics* 9 (1): 64–72. <u>https://doi.org/10.1375/twin.9.1.64</u>
- Bulik-Sullivan, Brendan, Hilary K. Finucane, Verneri Anttila, Alexander Gusev, Felix R. Day, Po Ru Loh, Laramie Duncan, et al. 2015. "An Atlas of Genetic Correlations across Human Diseases and Traits." *Nature Genetics* 47 (11): 1236–41. <u>https://doi.org/10.1038/ng.3406</u>
- Cawley, John, Davide Dragone, and Stephanie Von Hinke Kessler Scholder. 2016. "The Demand for Cigarettes as Derived from the Demand for Weight Loss: A Theoretical and Empirical Investigation." *Health Economics (United Kingdom)* 25 (1): 8–23. https://doi.org/10.1002/hec.3118
- Connett, J., J. Kusek, W. Bailey, P. O'Hara, and M. Wu. 1993. "Design of the Lung Health Study: A Randomized Clinical Trial of Early Intervention for Chronic Obstructive Pulmonary Disease." *Controlled Clinical Trials* 14: 3S-19S. https://pubmed.ncbi.nlm.nih.gov/8500311/
- Courtemanche, Charles, Rusty Tchernis, and Benjamin Ukert. 2018. "The Effect of Smoking on Obesity: Evidence from a Randomized Trial." *Journal of Health Economics* 57 (January):31–44. <u>https://doi.org/10.1016/j.jhealeco.2017.10.006</u>
- Cutler, D., and A. Lleras-Muney. 2006. Education and Health: Evaluating Theories and Evidence. NBER Working Paper 12352. https://www.nber.org/system/files/working_papers/w12352/w12352.pdf
- Daetwyler, Hans D., Beatriz Villanueva, and John A. Woolliams. 2008. "Accuracy of Predicting the Genetic Risk of Disease Using a Genome-Wide Approach." *PLoS ONE* 3 (10). <u>https://doi.org/10.1371/journal.pone.0003395</u>
- Duncan, Laramie E., and Matthew C. Keller. 2011. "A Critical Review of the First 10 Years of Candidate Gene-by-Environment Interaction Research in Psychiatry." *Am J Psychiatry* 168:1041–49. <u>https://psychiatryonline.org/doi/full/10.1176/appi.ajp.2011.11020191</u>
- Eldjarn, Grimur Hjorleifsson, Egil Ferkingstad, Sigrun H. Lund, Hannes Helgason, Olafur Th. Magnusson, Kristbjorg Gunnarsdottir, Thorunn A. Olafsdottir, et al. 2023. "Large-Scale Plasma Proteomics Comparisons through Genetics and Disease Associations." *Nature* 622 (7982): 348–58. <u>https://doi.org/10.1038/s41586-023-06563-x</u>
- Fahed, Akl C., Anthony A. Philippakis, and Amit V. Khera. 2022. "The Potential of Polygenic Scores to Improve Cost and Efficiency of Clinical Trials." *Nature Communications*. Nature Research. <u>https://doi.org/10.1038/s41467-022-30675-z</u>
- Fletcher, Jason M. 2012. "Why Have Tobacco Control Policies Stalled? Using Genetic Moderation to Examine Policy Impacts." *PLoS ONE* 7 (12). https://doi.org/10.1371/journal.pone.0050576
- Fletcher, Jason M., and Qiongshi Lu. 2021. "Health Policy and Genetic Endowments: Understanding Sources of Response to Minimum Legal Drinking Age Laws." *Health Economics* 30 (1): 194–203. <u>https://doi.org/10.1002/hec.4183</u>

- Fletcher, Jason, and Ryne Marksteiner. 2017. "Causal Spousal Health Spillover Effects and Implications for Program Evaluation." *American Economic Journal: Economic Policy* 9 (4): 144–66. <u>https://doi.org/10.1257/pol.20150573</u>
- Freedman, D., L. Khan, M. Serdula, D. Galuska, and W. Dietz. 2002. "Trends and Correlates of Class 3 Obesity in the United States From 1990 Through 2000." JAMA 288 (14): 1758– 1761. <u>https://jamanetwork.com/journals/jama/fullarticle/195356</u>
- Furberg, Helena, Yunjung Kim, Jennifer Dackor, Eric Boerwinkle, Nora Franceschini, Diego Ardissino, Luisa Bernardinelli, et al. 2010. "Genome-Wide Meta-Analyses Identify Multiple Loci Associated with Smoking Behavior." *Nature Genetics* 42 (5): 441–47. <u>https://doi.org/10.1038/ng.571</u>
- Galama, T., A. Lleras-Muney, and H. Kippersluis. 2018. The Effect of Education on Health and Mortality: A Review of Experimental and Quasi-Experimental Evidence. *NBER Working Paper 24225*. <u>https://www.nber.org/system/files/working_papers/w24225/w24225.pdf</u>
- Ge, Tian, Chia Yen Chen, Yang Ni, Yen Chen Anne Feng, and Jordan W. Smoller. 2019.
 "Polygenic Prediction via Bayesian Regression and Continuous Shrinkage Priors." *Nature Communications* 10 (1). <u>https://doi.org/10.1038/s41467-019-09718-5</u>
- Gibson, Greg. 2012. "Rare and Common Variants: Twenty Arguments." *Nature Reviews Genetics* 13: 135–145 <u>https://doi.org/10.1038/nrg3118</u>
- Grasing, Kenneth. 2016. "A Threshold Model for Opposing Actions of Acetylcholine on Reward Behavior: Molecular Mechanisms and Implications for Treatment of Substance Abuse Disorders." *Behavioural Brain Research* 312(1): 148-162 https://doi.org/10.1016/j.bbr.2016.06.022
- Gudmundsson, S., M. Singer-Berk, N. Watts, W. Phu, J. Goodrich, M. Solomonson, H. Rehm, D. MacArthur, and A. O'Donnell-Luria. 2022. "Variant Interpretation Using Population Databases: Lessons from gnomAD." *Human Mutation* 43 (8): 1012–1030. <u>https://doi.org/10.1002/humu.24309</u>
- Hansel, Nadia N., Ingo Ruczinski, Nicholas Rafaels, Don D. Sin, Denise Daley, Alla Malinina, Lili Huang, et al. 2013. "Genome-Wide Study Identifies Two Loci Associated with Lung Function Decline in Mild to Moderate COPD." *Human Genetics* 132 (1): 79–90. <u>https://doi.org/10.1007/s00439-012-1219-6</u>
- Heckman, J. J. 2001. "Micro Data, Heterogeneity, and the Evaluation of Public Policy: Nobel Lecture." *Journal of Political Economy* 109 (4): 673–748. <u>https://doi.org/10.1086/322086</u>
- Hellwege, Jacklyn N., Jacob M. Keaton, Ayush Giri, Xiaoyi Gao, Digna R. Velez Edwards, and Todd L. Edwards. 2017. "Population Stratification in Genetic Association Studies." *Current Protocols in Human Genetics* 95 (1): 1.22.1-1.22.23. <u>https://doi.org/10.1002/cphg.48</u>
- Howe, Laurence J., Michel G. Nivard, Tim T. Morris, Ailin F. Hansen, Humaira Rasheed, Yoonsu Cho, Geetha Chittoor, et al. 2022. "Within-Sibship Genome-Wide Association Analyses Decrease Bias in Estimates of Direct Genetic Effects." *Nature Genetics* 54 (5): 581–92. <u>https://doi.org/10.1038/s41588-022-01062-7</u>

- Kanner, R. E., N. R. Anthonisen, and J. E. Connett. 2001. "Lower Respiratory Illnesses Promote FEV(1) Decline in Current Smokers but Not Ex-Smokers with Mild Chronic Obstructive Pulmonary Disease: Results from the Lung Health Study." *American Journal of Respiratory and Critical Care Medicine* 164 (3): 358–364. https://doi.org/10.1164/ajrccm.164.3.2010017
- Lewis, Cathryn M., and Evangelos Vassos. 2020. "Polygenic Risk Scores: From Research Tools to Clinical Instruments." *Genome Medicine* 12(44). <u>https://doi.org/10.1186/s13073-020-00742-5</u>
- Lillard, D. 2017. "Educational Heterogeneity in the Association Between Smoking Cessation and Health Information." In Human Capital and Health Behavior (Advances in Health Economics and Health Services Research), edited by K. Bolin, B. Lindgren, M. Grossman, D. Gyrd-Hansen, T. Iversen, R. Kaestner, and J. Sindelar, 25: Chapter 6. *Emerald Publishing Limited*. <u>https://www.emerald.com/insight/content/doi/10.1108/S0731-</u> 219920170000025006/full/html
- Liu, Mengzhen, Yu Jiang, Robbee Wedow, Yue Li, David M. Brazel, Fang Chen, Gargi Datta, et al. 2019. "Association Studies of up to 1.2 Million Individuals Yield New Insights into the Genetic Etiology of Tobacco and Alcohol Use." *Nature Genetics* 51: 237–244. <u>https://doi.org/10.1038/s41588-018-0307-5</u>
- Manolio, Teri A., Francis S. Collins, Nancy J. Cox, David B. Goldstein, Lucia A. Hindorff, David J. Hunter, Mark I. McCarthy, et al. 2009. "Finding the Missing Heritability of Complex Diseases." *Nature* 461:747–753. <u>https://doi.org/10.1038/nature08494</u>
- Martin, Alicia R., Masahiro Kanai, Yoichiro Kamatani, Yukinori Okada, Benjamin M. Neale, and Mark J. Daly. 2019. "Clinical Use of Current Polygenic Risk Scores May Exacerbate Health Disparities." *Nature Genetics* 51 (4): 584–91. <u>https://doi.org/10.1038/s41588-019-0379-x</u>
- Meckel, Katherine, and Katherine P Rittenhouse. 2022. "The Effect of Smoking on Mental Health: Evidence from a Randomized Trial." *NBER Working Paper 29867*. <u>http://www.nber.org/papers/w29867</u>
- Miao, Jiacheng, Hanmin Guo, Gefei Song, Zijie Zhao, Lin Hou, and Qiongshi Lu. 2023. "Quantifying Portable Genetic Effects and Improving Cross-Ancestry Genetic Prediction with GWAS Summary Statistics." *Nature Communications* 14 (1). https://doi.org/10.1038/s41467-023-36544-7
- Nagel, Mats, Philip R. Jansen, Sven Stringer, et al. 2018. "Meta-analysis of genome-wide association studies for neuroticism in 449,484 individuals identifies novel genetic loci and pathways." *Nat Genet* 50, 920–927. <u>https://doi.org/10.1038/s41588-018-0151-7</u>
- Office of the Surgeon General. 2020. "Smoking Cessation: A Report of the Surgeon General." Washington, DC: U.S. Department of Health and Human Services. <u>https://www.hhs.gov/sites/default/files/2020-cessation-sgr-full-report.pdf</u>

- O'Hara, P., J. Grill, M. A. Rigdon, J. E. Connett, G. A. Lauger, and J. J. Johnston. 1993. "Design and Results of the Initial Intervention Program for the Lung Health Study." *Preventive Medicine* 22 (3): 304–315. <u>https://doi.org/10.1006/pmed.1993.1025</u>
- Pampel, Fred C., Patrick M. Krueger, and Justin T. Denney. 2010. "Socioeconomic Disparities in Health Behaviors." *Annual Review of Sociology* 36:349–70. <u>https://doi.org/10.1146/annurev.soc.012809.102529</u>
- Parker, Margaret M., Sharon M. Lutz, Brian D. Hobbs, Robert Busch, Merry Lynn N. McDonald, Peter J. Castaldi, Terri H. Beaty, John E. Hokanson, Edwin K. Silverman, and Michael H. Cho. 2019. "Assessing Pleiotropy and Mediation in Genetic Loci Associated with Chronic Obstructive Pulmonary Disease." *Genetic Epidemiology* 43 (3): 318–29. <u>https://doi.org/10.1002/gepi.22192</u>
- Price, Alkes L., Nick J. Patterson, Robert M. Plenge, Michael E. Weinblatt, Nancy A. Shadick, and David Reich. 2006. "Principal Components Analysis Corrects for Stratification in Genome-Wide Association Studies." *Nature Genetics* 38 (8): 904–9. <u>https://doi.org/10.1038/ng1847</u>
- Purcell, Shaun, Benjamin Neale, Kathe Todd-Brown, Lori Thomas, Manuel A R Ferreira, David Bender, Julian Maller, Pamela Sklar, Paul I W De Bakker, et al. 2007. "PLINK: A Tool Set for Whole-Genome Association and Population-Based Linkage Analyses." *American Journal of Human Genetics* 81 (3): 559–75. <u>https://doi.org/10.1086/519795</u>
- Ruan, Y., Y. Lin, Y. Feng, C. Chen, M. Lam, Z. Guo, S. Initiatives, L. He, A. Sawa, A. Martin, S. Qin, H. Huang, and T. Ge. 2022. "Improving Polygenic Prediction in Ancestrally Diverse Populations." *Nature Genetics* 54 (5): 573–580. https://pmc.ncbi.nlm.nih.gov/articles/PMC9117455/
- Schmitz, Lauren L., and Dalton Conley. 2017. "The Effect of Vietnam-Era Conscription and Genetic Potential for Educational Attainment on Schooling Outcomes." *Economics of Education Review* 61 (December):85–97. <u>https://doi.org/10.1016/j.econedurev.2017.10.001</u>
- Schmitz, Lauren L., Arianna M. Gard, and Erin B. Ware. 2019. "Examining Sex Differences in Pleiotropic Effects for Depression and Smoking Using Polygenic and Gene-Region Aggregation Techniques." *American Journal of Medical Genetics, Part B: Neuropsychiatric Genetics* 180 (6): 448–68. https://doi.org/10.1002/ajmg.b.32748
- Panagiotou, Orestis A., Ewoud Schuit, Marcus R. Munafò, Derrick A. Bennett, Andrew W. Bergen, and Sean P. David. 2019. "Smoking Cessation Pharmacotherapy Based on Genetically-Informed Biomarkers: What is the Evidence?" Nicotine & Tobacco Research 21 (9): 1289–93. https://academic.oup.com/ntr/article/21/9/1289/5301290
- Shrine, Nick, Anna L. Guyatt, A. Mesut Erzurumluoglu, Victoria E. Jackson, Brian D. Hobbs, Carl A. Melbourne, Chiara Batini, et al. 2019. "New Genetic Signals for Lung Function Highlight Pathways and Chronic Obstructive Pulmonary Disease Associations across Multiple Ancestries." *Nature Genetics* 51 (3): 481–93. <u>https://doi.org/10.1038/s41588-018-0321-7</u>

- Shrine, Nick, Abril G. Izquierdo, Jing Chen, Richard Packer, Robert J. Hall, Anna L. Guyatt, Chiara Batini, et al. 2023. "Multi-Ancestry Genome-Wide Association Analyses Improve Resolution of Genes and Pathways Influencing Lung Function and Chronic Obstructive Pulmonary Disease Risk." *Nature Genetics* 55 (3): 410–22. <u>https://doi.org/10.1038/s41588-023-01314-0</u>
- Slob, Eric A.W., and Cornelius A. Rietveld. 2021. "Genetic Predispositions Moderate the Effectiveness of Tobacco Excise Taxes." *PLoS ONE* 16 (11 November). <u>https://doi.org/10.1371/journal.pone.0259210</u>
- Solovieff, Nadia, Chris Cotsapas, Phil H. Lee, Shaun M. Purcell, and Jordan W. Smoller. 2013. "Pleiotropy in Complex Traits: Challenges and Strategies." *Nature Reviews Genetics* 14: 483–495. <u>https://doi.org/10.1038/nrg3461</u>
- Thorgeirsson, T. E., D. F. Gudbjartsson, P. Sulem, S. Besenbacher, U. Styrkarsdottir, G. Thorleifsson, G. B. Walters, et al. 2013. "A Common Biological Basis of Obesity and Nicotine Addiction." *Translational Psychiatry* 3:1–7. <u>https://doi.org/10.1038/tp.2013.81</u>
- Torkamani, Ali, Nathan E. Wineinger, and Eric J. Topol. 2018. "The Personal and Clinical Utility of Polygenic Risk Scores." *Nature Reviews Genetics* 19: 581–590. <u>https://doi.org/10.1038/s41576-018-0018-x</u>
- Vink, Jacqueline M., Gonneke Willemsen, and Dorret I. Boomsma. 2005. "Heritability of Smoking Initiation and Nicotine Dependence." *Behavior Genetics* 35 (4): 397–406. <u>https://doi.org/10.1007/s10519-004-1327-8</u>
- Wang, Tao, Jee Young Moon, Yiqun Wu, Christopher I. Amos, Rayjean J. Hung, Adonina Tardon, Angeline Andrew, et al. 2017. "Pleiotropy of Genetic Variants on Obesity and Smoking Phenotypes: Results from the Oncoarray Project of the International Lung Cancer Consortium." *PLoS ONE* 12 (9). <u>https://doi.org/10.1371/journal.pone.0185660</u>
- Wedow, Robbee, Meghan Zacher, Brooke M Huibregtse, Kathleen Mullan Harris, Benjamin W Domingue, and Jason D Boardman. 2018. "Education, Smoking, and Cohort Change: Forwarding a Multidimensional Theory of the Environmental Moderation of Genetic Effects." *American Sociological Review* 83 (4): 802–32. https://doi.org/10.1177/0003122418785368
- Wray, Naomi R., Kathryn E. Kemper, Benjamin J. Hayes, Michael E. Goddard, and Peter M. Visscher. 2019. "Complex Trait Prediction from Genome Data: Contrasting EBV in Livestock to PRS in Humans." *Genetics* 211 (4): 1131–41. https://doi.org/10.1534/genetics.119.301859
- Wu, Y., X. Zhong, Y. Lin, Z. Zhao, J. Chen, B. Zheng, J. Li, J. Fletcher, and Q. Lu. 2021.
 "Estimating Genetic Nurture with Summary Statistics of Multigenerational Genome-Wide Association Studies." PNAS 118 (25).
 https://www.pnas.org/doi/full/10.1073/pnas.2023184118
- Xian, H., J. F. Scherrer, P. A. F. Madden, M. J. Lyons, M. Tsuang, W. R. True, and S. A. Eisen. 2003. "The Heritability of Failed Smoking Cessation and Nicotine Withdrawal in Twins

Who Smoked and Attempted to Quit." *Nicotine & Tobacco Research* 5 (2): 245–54. <u>https://doi.org/10.1080/1462220031000073667</u>

- Xu, Ke, Boyang Li, Kathleen A. McGinnis, Rachel Vickers-Smith, Cecilia Dao, Ning Sun, Rachel L. Kember, et al. 2020. "Genome-Wide Association Study of Smoking Trajectory and Meta-Analysis of Smoking Status in 842,000 Individuals." *Nature Communications* 11 (1). <u>https://doi.org/10.1038/s41467-020-18489-3</u>
- Yang, Jian, Jian Zeng, Michael E. Goddard, Naomi R. Wray, and Peter M. Visscher. 2017.
 "Concepts, Estimation and Interpretation of SNP-Based Heritability." *Nature Genetics* 49: 1304–1310. <u>https://doi.org/10.1038/ng.3941</u>
- Yengo, Loic, Julia Sidorenko, Kathryn E. Kemper, Zhili Zheng, Andrew R. Wood, Michael N. Weedon, Timothy M. Frayling, Joel Hirschhorn, Jian Yang, and Peter M. Visscher. 2018.
 "Meta-Analysis of Genome-Wide Association Studies for Height and Body Mass Index in ~700 000 Individuals of European Ancestry." *Human Molecular Genetics* 27 (20): 3641–49. https://doi.org/10.1093/hmg/ddy271
- Young, Alexander I. 2019. "Solving the Missing Heritability Problem." *PLoS Genetics* 15 (6). <u>https://doi.org/10.1371/journal.pgen.1008222</u>
- Young, Alexander I., Stefania Benonisdottir, Molly Przeworski, and Augustine Kong. 2019.
 "Deconstructing the Sources of Genotype-Phenotype Associations in Humans." Science 365 (6460): 1396–1400. <u>https://doi.org/10.1126/science.aax3710</u>
- Zheng, Jie, A. Mesut Erzurumluoglu, Benjamin L. Elsworth, John P. Kemp, Laurence Howe, Philip C. Haycock, Gibran Hemani, et al. 2017. "LD Hub: A Centralized Database and Web Interface to Perform LD Score Regression That Maximizes the Potential of Summary Level GWAS Data for SNP Heritability and Genetic Correlation Analysis." *Bioinformatics* 33 (2): 272–79. https://doi.org/10.1093/bioinformatics/btw613

SUPPLEMENTAL APPENDICES

Understanding the Role of Genetic Heterogeneity in Smoking Interventions: Experimental Evidence from the Lung Health Study

Shubhashrita Basu Jason Fletcher Qiongshi Lu Jiacheng Miao Lauren L. Schmitz

APPENDIX A

List of Tables

Table A1.	Mean Baseline Characteristics of the LHS Sample by Genotyped Status 1	
Table A2.	Impact of Treatment Status on the Probability of Being Genotyped in the LHS 2	
Table A3.	Incremental R ² Values of PGIs and Demographic Characteristics	;
Table A4.	Heterogeneous Treatment Effects of the LHS Intervention on Smoking Cessation by SC PGI	ł
Table A5.	Heterogeneous Treatment Effects of the LHS Intervention on Smoking Cessation by Baseline CPD and SI PGI	;
Table A6.	Heterogeneous Treatment Effects of the LHS Intervention on Smoking Cessation by SI PGI and Annual Visit	
Table A7.	Heterogeneous Treatment Effects of the LHS Intervention on Cigarettes Per Day by CPD PGI and Annual Visit	,

List of Figures

Figure A1. PGI Correlation Matrix in the LHS Sample	8
Figure A2. Mean CPD at Baseline by CPD PGI Quintile	9
Figure A3. Mean BMI at Baseline by BMI PGI Quintile 1	0
Figure A4. Mean FEV ₁ /FVC at Baseline by FEV ₁ /FVC PGI Quintile	11
Figure A5. Mean Depression Symptoms at Baseline by Depression PGI Quintile 1	12
Figure A6. Per-wave Means of Smoking Cessation and CPD by Treatment Status	13
Figure A7. Heterogeneous Treatment Effects of the LHS Intervention on Smoking Cessation by Baseline CPD and SI PGI	14

APPENDIX B

B1. List of Abbreviations	15
B2. Additional Information on the Lung Health Study Intervention	16
B3. Quality Control and Imputation of the Genotyped Data	18

Appendix A

Variables	Not Genotyped	Genotyped	p-value of difference in means
Body mass index (BMI)	25.650	25.529	0.280
Cigarettes per day (CPD)	31.328	31.249	0.830
FEV ₁ /FVC	62.923	62.943	0.899
Age	48.306	48.537	0.236
Pack years	40.127	40.586	0.401
Years of education	13.598	13.614	0.844
Female	0.375	0.370	0.703
Ν	1,742	4,145	

 Table A1. Mean Baseline Characteristics of the Lung Health Study (LHS)
 Sample by Genotyped Status

Note: The table reports average respondent background characteristics for LHS participants stratified by genotyped status. All variables were measured at baseline before the treatment was administered. We additionally report the p-values of the differences in means for each variable.

	Genotyped
Treated	0.001
	(0.013)
N	5.007
	5,887

 Table A2. Impact of Treatment Status on the Probability of Being

 Genotyped in the Lung Health Study (LHS)

Note: The table reports the effect of treatment on the probability of being genotyped in the LHS. The total number of participants is 5,887. The regression controls for age fixed effects and sex. Robust standard errors are reported in parentheses.

PGIs	Smoking cessation (all waves)	CPD (baseline)	BMI (baseline)	FEV ₁ /FVC (baseline)
SC PGI	0.19%			
Bootstrapped SE	(0.056)			
p-value	0.002			
SI PGI	0.03%			
Bootstrapped SE	(0.02)			
p-value	0.242			
CPD PGI		3.33%		
Bootstrapped SE		(0.55)		
p-value		0.000		
BMI PGI			8.82%	
Bootstrapped SE			(0.84)	
p-value			0.000	
FEV ₁ /FVC PGI				0.01%
Bootstrapped SE				(0.35)
p-value				0.000
Demographic characteristics				
Sex	0.07%	2.15%	8.39%	0.26%
Bootstrapped SE	(0.037)	(0.433)	(0.83)	(0.16)
p-value	0.053	0.000	0.000	0.097
Married	0.17%	0.03%	0.87%	0.01%
Bootstrapped SE	(0.056)	(0.062)	(0.31)	(0.05)
p-value	0.003	0.639	0.006	0.788

Table A3. Incremental R² Values of Polygenic Indices and Demographic Characteristics

Note: The table reports the incremental R^2 values of PGIs and demographic characteristics for smoking cessation (SC), smoking initiation (SI), cigarettes per day (CPD), body mass index (BMI), and lung function, or forced expiratory volume in one second over forced vital capacity (FEV₁/FVC). For all the variables except smoking cessation, we use baseline measures (data on smoking cessation is not available before the treatment). Each estimate comes from a regression of the outcome variable on the PGI or demographic characteristic after residualizing out the first 20 genetic principal components (PCs) to account for population stratification. Bootstrapped standard errors with 1000 replications and corresponding p-values are reported in parentheses.

	2				
	Composite	SIA	SIP		
	(1)	(2)	(3)		
	Smoking Cessation				
Treated	0.229	0.232	0.227		
	(0.011)	(0.014)	(0.014)		
SC	-0.015	-0.013	-0.017		
	(0.008)	(0.009)	(0.008)		
Treat × SC PGI (std)	-0.009	-0.002	-0.014		
	(0.011)	(0.014)	(0.013)		
Ν	20,725	13,950	13,690		

Table A4. Heterogeneous Treatment Effects of the LHS Intervention on
Smoking Cessation by SC PGI

Note: The table reports the heterogeneous effects of the treatment on the probability of quitting smoking. Composite indicates both treatment arms combined. SIA: smoking intervention plus Atrovent inhaler group. SIP: smoking intervention plus placebo inhaler group. The PGI is standardized. All the regressions control for age fixed effects, wave fixed effects, sex, and the first 20 principal components (PCs) of the genetic data to account for population stratification. Standard errors clustered at the individual level are reported in the parentheses.

	Composite	SIA	SIP		
Smoking Cessation					
Panel I					
Treated	0.229	0.230	0.227		
	(0.011)	(0.014)	(0.014)		
Baseline CPD (std)	-0.029	-0.030	-0.028		
	(0.009)	(0.009)	(0.009)		
Treat × Baseline CPD (std)	0.012	0.009	0.017		
	(0.012)	(0.014)	(0.015)		
Panel II					
Treated	0.231	0.233	0.228		
	(0.011)	(0.014)	(0.014)		
Baseline CPD (std)	-0.028	-0.029	-0.028		
	(0.009)	(0.009)	(0.009)		
Treat × Baseline CPD (std)	0.013	0.010	0.017		
	(0.012)	(0.014)	(0.015)		
SI PGI (std)	0.004	0.005	0.002		
	(0.008)	(0.008)	(0.008)		
Treat × SI PGI (std)	-0.025	-0.034	-0.013		
	(0.011)	(0.013)	(0.013)		
Ν	20,725	13,950	13,690		

Table A5. Heterogeneous Treatment Effects of the LHS Intervention on Smoking Ce	ssation
by Baseline CPD and SI PGI	

Note: Panel I of this table reports heterogeneous treatment effects of cigarettes smoked per day (CPD) at baseline on smoking cessation. These regressions control for age fixed effects, wave fixed effects, and sex. Panel II adds the smoking initiation (SI) PGI and its interaction. Regressions in Panel II also control for the first 20 principal components (PCs) of the genetic data to account for population stratification. Standard errors clustered at the individual level are reported in parentheses.

	Annual Visit				
	1	2	3	4	5
Panel 1	Composite				
Treated	0.288	0.240	0.235	0.205	0.183
	(0.012)	(0.013)	(0.014)	(0.014)	(0.015)
SI PGI (std)	0.002	0.007	0.011	-0.003	0.002
	(0.007)	(0.009)	(0.010)	(0.011)	(0.012)
Treat × SI PGI (std)	-0.021	-0.020	-0.033	-0.014	-0.033
	(0.012)	(0.013)	(0.013)	(0.014)	(0.015)
Ν	4,145	4,145	4,145	4,145	4,145
Panel II			SIA		
Treated	0.296	0.243	0.232	0.204	0.187
	(0.015)	(0.016)	(0.016)	(0.017)	(0.017)
SI PGI (std)	0.003	0.007	0.011	-0.001	0.002
	(0.008)	(0.009)	(0.010)	(0.011)	(0.012)
Treat \times SI PGI (std)	-0.031	-0.032	-0.038	-0.027	-0.043
	(0.015)	(0.016)	(0.016)	(0.017)	(0.017)
Ν	2,790	2,790	2,790	2,790	2,790
Panel III			SIP		
Treated	0.278	0.235	0.236	0.205	0.178
	(0.015)	(0.016)	(0.017)	(0.017)	(0.018)
SI PGI (std)	-0.001	0.004	0.008	-0.007	-0.000
	(0.008)	(0.009)	(0.010)	(0.011)	(0.012)
Treat \times SI PGI (std)	-0.010	-0.004	-0.026	0.003	-0.020
	(0.015)	(0.016)	(0.016)	(0.017)	(0.018)
Ν	2.738	2.738	2.738	2.738	2.738

 Table A6. Heterogeneous Treatment Effects of the LHS Intervention on Smoking Cessation by

 Smoking Initiation PGI and Annual Visit

Note: This table reports the heterogeneous treatment effects of the SI PGI on smoking cessation stratified by annual visits post treatment. There were five follow-up visits recorded in the LHS. Panel I reports the effects for both treatment groups combined (SIA and SIP) relative to the control group. Panel II reports the effects of the SIP participants who were assigned to the SIA group compared to the control group. Panel III reports the effects of the SIP participants in comparison with the control group. Each regression controls for age fixed effects, wave fixed effects, sex, and the first 20 principal components to account for population stratification.

	51.0 1 610				
	Annual Visit				
	1	2	3	4	5
Panel 1	Composite				
Treated	-11.995	-9.850	-9.330	-7.727	-6.626
	(0.491)	(0.486)	(0.484)	(0.488)	(0.484)
CPD PGI (std)	1.814	1.544	1.872	1.673	1.665
	(0.437)	(0.429)	(0.427)	(0.436)	(0.441)
Treat × CPD PGI (std)	-0.961	-0.313	-0.799	-0.485	-0.560
	(0.514)	(0.512)	(0.506)	(0.513)	(0.515)
N	4,050	4,038	4,027	4,019	4,145
Panel II			SIA		
Treated	-12.319	-9.995	-9.657	-7.906	-6.935
	(0.565)	(0.559)	(0.550)	(0.557)	(0.550)
CPD PGI (std)	1.842	1.503	1.861	1.601	1.623
	(0.437)	(0.432)	(0.430)	(0.440)	(0.446)
Treat × CPD PGI (std)	-1.724	-0.824	-1.153	-0.805	-0.681
	(0.580)	(0.587)	(0.577)	(0.582)	(0.578)
Ν	2,727	2,719	2,708	2,700	2,790
Panel III			SIP		
Treated	-11.531	-9.589	-8.895	-7.522	-6.248
	(0.575)	(0.570)	(0.572)	(0.571)	(0.563)
CPD PGI (std)	1.886	1.637	1.937	1.761	1.741
	(0.438)	(0.434)	(0.431)	(0.440)	(0.444)
Treat × CPD PGI (std)	-0.174	0.184	-0.487	-0.182	-0.460
	(0.591)	(0.592)	(0.587)	(0.590)	(0.591)
Ν	2,675	2,671	2,657	2,663	2,738

 Table A7. Heterogeneous Treatment Effects of the LHS Intervention on Cigarettes per Day by CPD PGI and Annual Visit

Note: This table reports the heterogeneous treatment effect of the CPD PGI on CPD stratified by annual visits post treatment. There were five follow-up visits recorded in the LHS. Panel I reports effects for both treatment groups combined (SIA and SIP) relative to the control group. Panel II reports effects for participants who were assigned to the SIA group compared to the control group. Panel III reports effects for SIP participants compared to the control group. Each regression controls for age fixed effects, wave fixed effects, sex, and the first 20 principal components (PCs) of the genetic data to account for population stratification.

Figures



Figure A1. PGI Correlation Matrix in the LHS Sample

Note: The heatmap presents the correlation coefficients between the following PGIs: smoking initiation (SI), smoking cessation (SC), cigarettes per day (CPD), body mass index (BMI), lung function, or forced expiratory volume in one second over forced vital capacity (FEV_1/FVC), and depression (DP). Lighter cells indicate weaker correlations, while darker cells indicate stronger correlations.



Figure A2. Mean Cigarettes Per Day at Baseline by CPD PGI Quintile

Note: The figure plots the mean number of cigarettes per day (CPD) at baseline by CPD PGI quintiles. 95% confidence intervals are reported.



Figure A3. Mean Body Mass Index at Baseline by BMI PGI Quintile

Note: This figure plots the mean body mass index (BMI) at baseline by BMI PGI quintiles. 95% confidence intervals are reported.



Figure A4. Mean FEV₁/FVC at Baseline by FEV₁/FVC PGI Quintile

Note: The figure plots mean lung function, or forced expiratory volume in one second over forced vital capacity (FEV₁/FVC) at baseline by FEV₁/FVC PGI quintiles. 95% confidence intervals are reported.

Figure A5. Mean Depression Symptoms at Baseline by Depression PGI Quintile



Note: This figure plots the average distress index at baseline by depression PGI quintile. The distress index is the sum of five symptoms related to depression: irritability, insomnia, mood changes, nervousness, and psychological illness. The LHS asks participants about these mental conditions via the following question: "Indicate the extent to which you have been troubled in the last four months by any of the following. Please indicate 'Severe,' 'Moderate,' 'Mild,' or 'Not at all.'" For each condition, we construct a dichotomous variable that takes the value of one if the respondent reports mild to severe symptoms and zero otherwise. We then form the distress index by summing across these mental health conditions at baseline. An increase in the distress index indicates worsening mental health. 95% confidence intervals are reported.



Figure A6. Per-wave Means of Smoking Cessation and CPD by Treatment Status

Panel I

Panel II

Note: Panel I shows the averages of smoking cessation and CPD by treatment status at baseline before treatment was administered and across all five annual follow-up visits. SIA: smoking intervention plus Atrovent inhaler (ipratropium bromide) group; SIP: smoking intervention plus placebo inhaler group. Averages are shown for the two treatment arms separately, for the composite treatment group (SIA and SIP), and the control group. Since we do not have information on smoking cessation at baseline, the first diagram under Panel I starts at the first visit. Panel II presents a zoomed-in version of the Panel I figures for the three treatment groups. Since baseline CPD was the same across groups, Panel II only plots means for the five follow-up visits.

Figure A7. Heterogeneous Treatment Effects of the LHS Intervention on Smoking Cessation by Baseline CPD and SI PGI



Note: Panel I corresponds to the composite treatment group ('1' if either SIA or SIP and '0' if control), Panel II corresponds to the SIA treatment group ('1' if SIA and '0' if control), and Panel III corresponds to the SIP treatment group ('1' if SIP and '0' if control). The top row presents the heterogeneous treatment effects on smoking cessation using variations in baseline CPD. These regressions control for age fixed effects, wave fixed effects, and sex. The lower row plots predicted treatment effects for smoking cessation by SI PGI that also control for baseline CPD and its interaction with the treatment. These regressions also control for the first 20 PCs of the genetic data. The grey diamonds represent the treated groups (composite/SIA/SIP) and the black circles represent the control group. Standard errors are clustered at the individual level and 95% confidence intervals are plotted with the estimates.

Appendix B

B1. List of Abbreviations

- 1. LHS: Lung Health Study
- 2. RCT: Randomized controlled trial
- 3. ATE: Average treatment effect
- 4. SIA: Smoking intervention plus inhaler with the active drug ipratropium bromide group
- 5. SIP: Smoking intervention plus placebo inhaler group
- 6. PGI: Polygenic index
- 7. PC: Principal component
- 8. SI: Smoking initiation
- 9. SC: Smoking cessation
- 10. CPD: Cigarettes smoked per day
- 11. FEV₁/FVC: Forced expiratory volume in one second over forced vital capacity
- 12. DP: Depression
- 13. GWAS: Genome-wide association study
- 14. SNP: Single nucleotide polymorphism
- 15. LD: Linkage disequilibrium

B2. Additional Information on the Lung Health Study Intervention

1. Exclusionary criteria

Individuals were excluded from the study if they had serious illnesses such as cancer, heart disease, stroke, or other significant medical conditions like high blood pressure, were using bronchodilators, beta-blockers, nitrates, or insulin, or if they consumed more than 25 alcoholic drinks per week or were binge drinkers. To reduce attrition, individuals were also excluded if they anticipated moving more than 75 miles from the clinical center or were unwilling to participate in the behavior intervention if randomized into the treatment group.

2. Intervention details

Each center recruited approximately 600 participants and assigned 400 individuals to the treatment group and 200 to the control group. Women comprised approximately 37 percent of enrolled subjects and 96 percent of the sample was white. The smoking cessation intervention program was designed to help individuals cope with their addiction and included several key features throughout the five-year trial (Connett et al. 1993; O'Hara et al. 1993):

- An initial message from a physician was delivered soon after randomization to give the participant information regarding their lung impairment, the adverse health effects of smoking, the importance of smoking cessation to reduce health risks, and a prescription for the inhaler and nicotine gum. Participants were encouraged to limit their nicotine gum use to the first 6 months of the trial and were provided with a reduction strategy if they had difficulty tapering their use.
- A meeting with an intervention specialist (held immediately after the physician meeting) explained the program and helped participants choose a quit date.

- An intensive 12-session group intervention program spread over 10 weeks provided cognitive and behavioral strategies for quitting and instructions for proper inhaler and nicotine gum use.
- Clinic visits every four months across all five years of follow-up.
- A maintenance program to minimize relapse and provide long-term support with problems such as weight gain and stress management.
- An extended intervention program for smokers who relapsed included options for individual counseling, physician visits, or additional group meetings.

References

- Connett, J., J. Kusek, W. Bailey, P. O'Hara, and M. Wu. 1993. "Design of the Lung Health Study: A Randomized Clinical Trial of Early Intervention for Chronic Obstructive Pulmonary Disease." *Controlled Clinical Trials* 14: 3S-19S.
- O'Hara, P., J. Grill, M. A. Rigdon, J. E. Connett, G. A. Lauger, and J. J. Johnston. 1993. "Design and Results of the Initial Intervention Program for the Lung Health Study." *Preventive Medicine* 22 (3): 304–315.

B3. Quality Control and Imputation of the Genotyped Data

We performed pre-imputation quality control (QC) on the genotype data using PLINK software (Purcell et al. 2007). We only kept autosomal non-biallelic SNPs with a minor allele frequency (MAF) > 0.01 and Hardy Weinberg equilibrium test p-value \geq 1.0e-6. We used the UCSC liftOver tool to lift over the genome coordinates from hg18 to hg19. We phased and imputed the genotype data using the HRC reference panel version r1.1 2016 from the Michigan Imputation server (Das et al. 2016). After imputation, we removed the duplicated and strand ambiguous SNPs, SNPs with imputation quality<0.9, and SNPs with MAF<0.01. After QC, 12,030,369 SNPs remained. PGIs were then constructed using SNPs that overlapped between the LHS and GWAS samples. Number of SNPs used to construct each PGI: SI=1,044,939; SC=1,057,529; CPD=1,057,519; BMI=920,776; FEV1/FVC=1,052,022; Depression=1,017,199.

References

- Das, Sayantan, Lukas Forer, Sebastian Schönherr, Carlo Sidore, Adam E Locke, Alan Kwong, Scott I Vrieze, et al. 2016. "Next-Generation Genotype Imputation Service and Methods." *Nature Genetics* 48 (10): 1284–87.
- Purcell, Shaun, Benjamin Neale, Kathe Todd-Brown, Lori Thomas, Manuel A R Ferreira, David Bender, Julian Maller, Pamela Sklar, Paul I W De Bakker, et al. 2007. "PLINK: A Tool Set for Whole-Genome Association and Population-Based Linkage Analyses." *American Journal of Human Genetics* 81 (3): 559–75.